\begin{table}%t1
\caption{\label{tabledata}Basic data for the program stars, ordered by increasing right ascension. Projected
rotational velocities are indicated for Be stars, while the presence of a circumstellar nebula is indicated for LBVs and related objects.
References are: CN04~=~\citet{cn04};
Cro06~=~\citet{crowther06}; Mir01~=~\citet{mir01};
L97~=~\citet{l97}; LL06~=~\citet{ll06}; Riv97~=~\citet{rivinius97}; R97~=~\citet{roche97}; 
S06~=~\citet{s06}; Sta03~=~\citet{stahl03};
vG92~=~\citet{vg92}; vG01~=~\citet{vg01};
WF2000~=~\citet{wf2000}; Z05~=~\citet{zorec05}. }
\small%\centering
\par
\begin{tabular}{c c c c c c c c c}
\multicolumn{9}{c}{Be stars} \\
\hline\hline\\[-3mm]
Name & HD & RA (J2000) & Dec (J2000)  & Spectral type & $v~\sin{{i}}$ (km s$^{-1}$)& Reference & Fig.& Obs. Date \\
\hline
X Per   & 24534  & 03 55 23.1 & +31 02 45.0       & B0Ve   & 293 & L97, R97, Z05  &~\ref{be3} & 02 Nov. 04  \\
$\beta$ Mon~A   & 45725  & 06 28 49.4 & --07 02 03.5 & B3Ve   & 330 & Z05      &~\ref{be3}     & 02 Nov. 04 \\
$\delta$~Cen  & 105435 & 12 08 21.5 & --50 43 20.7    & B2Vne  & 240 & LL06     &~\ref{be2}     & 01 Jun. 09 \\
HD 110432  & 110432 & 12 42 50.3 & --63 03 31.0       & B1IVe  & 300 & S06      &~\ref{be3}     & 02 May 01 \\
HD 120991  & 120991 & 13 53 57.2 & --47 07 41.4       & B2Ve   & 130 & LL06     &~\ref{be2}     & 01 Jun. 09  \\
$\eta$ Cen    & 127972 & 14 35 30.4 & --42 09 28.2    & B2Ve   & 310 & LL06     &~\ref{be1}     & 01 Jun. 09   \\
HD 142983  & 142983 & 15 58 11.4 & --14 16 45.7       & B3IVe  & 390 & Z05      &~\ref{be1}     & 01 Jun. 09   \\
$\delta$~Sco  & 143275 & 16 00 20.0 & --22 37 18.2    & B0.3IVe & 150 & Mir01   &~\ref{be3}     & 02 Apr. 29  \\
$\chi$ Oph    & 148184 & 16 27 01.4 & --18 27 22.5    & B1.5Ve & 144 & Z05      &~\ref{be2}     & 01 Jun. 12      \\
$\alpha$ Ara   & 158427 & 17 31 50.5 & --49 52 34.1   & B2Vne  & 270 & LL06     &~\ref{be1}     & 01 Jun. 09    \\
66 Oph     & 164284 & 18 00 15.8 & +04 22 07.0       & B2Ve   & 220 & LL06     &~\ref{be1}     & 01 Jun. 09     \\
$\lambda$ Pav & 173948 & 18 52 13.0 & --62 11 15.3    & B2IIIe  & 140 & Z05     &~\ref{be3}     & 02 Apr. 29     \\
$\upsilon$~Cyg & 202904 & 21 17 55.1 & +34 53 48.8   & B2 Ve   & 185 & Z05     &~\ref{be2}     & 01 Jun. 10  	  \\
\hline   
&&&&&\\
\end{tabular}
\begin{tabular}{c c c c c c c c c}
\multicolumn{9}{c}{LBVs, LBV candidates and ex/dormant LBVs} \\
\hline\hline
Name & HD & RA (J2000) & Dec (J2000)  & Spectral type & Nebula  & Reference &  Fig. & Obs. Date(s) \\
\hline
HD 87643   & 87643  & 10 04 30.3 & --58 39 52.1      & Bpec	   & yes   & WF2000	 &~\ref{bpec1}  	    & 01 Jun. 12		  \\
HR~Car     & 90177  & 10 22 53.8 & --59 37 28.4      & B3Ia--A2Ia    & yes  & WF2000, vG01  &~\ref{lbv1},~\ref{var3}  & 01 Jun. 12, 04 Apr. 30   \\
$\eta$~Car & 93308  & 10 45 03.6 & --59 41 03.3        & Bpec	     & yes   & WF2000, vG01 &~\ref{lbv2},~\ref{var2} & 01 Jun. 12, 02 Nov. 04,  \\
&&&&&&&&  03 Jun. 28, 04 Apr. 30 \\
GG~Car     & 94878  & 10 55 58.0 & --60 23 33.4      & Bpec	     & no    & WF2000	   &~\ref{bpec1}	    & 01 Jun. 12	    \\
AG~Car     & 94910  & 10 56 11.6 & --60 27 12.8   & WN11h--Apec      & yes  & WF2000, vG01  &~\ref{lbv3},~\ref{var1}  & 01 Jun. 09, 02 Nov. 04,  \\
&&&&&&&& 04 Apr. 30, 04 Jul. 02 \\ 
He 3-519   & --      & 10 53 59.6 & --60 26 44.3   & WN11h	    & yes  & WF2000, vG01  &~\ref{lbv2},~\ref{var2}  & 01 Jun. 12, 04 Apr. 30  \\
W 243      & --      & 16 47 07.5 & --45 52 28.5       & A2Ia	     & no   & CN04  &~\ref{lbv1}		    & 04 Jul. 02	    \\
HD 326823  & 326823 & 17 06 53.9 & --42 36 39.7       & Bpec	     & no    & WF2000	    &~\ref{bpec1}	    & 01 Jun. 12	    \\
HD 160529  & 160529 & 17 41 59.0 & --33 30 13.7   & A2Ia 	     & no   & Sta03	   &~\ref{lbv1},~\ref{var3} & 01 Jun. 12, 04 Apr. 30   \\
HD 316285  & 316285 & 17 48 14.0 & --28 00 53.5   & Bpec 	     & no    & WF2000	   &~\ref{lbv2},~\ref{var2} & 01 Jun. 09, 02 Nov. 04,   \\
&&&&&&&& 04 Apr. 30 \\
HD 168607  & 168607 & 18 21 14.7 & --16 22 32.1   & B9Iape	     & no   & WF2000	   &~\ref{lbv1},~\ref{var3} & 01 Jun. 09, 04 Jul. 02   \\
HD 168625  & 168625 & 18 21 19.5 & --16 22 26.0       & B6Iap	     & yes  & WF2000	   &~\ref{lbv1},~\ref{var3} & 01 Jun. 09		 \\
MWC 314    & --      & 19 21 34.0 & +14 52 56.9   & Bpec 	     & no    & vG01, Mir03  &~\ref{lbv3},~\ref{var1} & 01 Jun. 09, 02 Nov. 04,   \\
&&&&&&&& 04 Apr. 30, 04 Jul. 02 \\
P~Cyg      & 193237 & 20 17 47.2 & +38 01 58.5   & Bpec 	     & yes   & WF2000, vG01 &\ref{lbv3},~\ref{var1}  & 01 Jun. 10, 02 Nov. 04,  \\
&&&&&&&& 04 Jul. 02\\  
\hline
&&&&&&&&\\
\end{tabular}
\begin{tabular}{c c c c c c c c}
\multicolumn{8}{c}{OB Supergiants} \\
\hline\hline
Name & HD & RA (J2000) & Dec (J2000)  & Spectral type & Reference & Fig. & Obs. Date(s) \\
\hline
HD 80077   & 80077  & 09 15 54.8 & --49 58 24.6         & B2Ia+  &  vG92, vG01	      &~\ref{obsuper}	  & 01 Jun. 12	      \\
HD 151804  & 151804 & 16 51 33.7 & --41 13 49.9        & O8Iaf	& WF2000    &~\ref{obsuper}		  & 01 Jun. 09	       \\
WR 79a     & 152408 & 16 54 58.5 & -41 09 03.1         & O8:Iafpe,WN9ha & WF2000	  &~\ref{obsuper} & 01 Jun. 09	       \\
$\zeta^1$ Sco  & 152236 & 16 53 59.7 & --42 21 43.3 & B1.5Ia+  & Cro06	&~\ref{obsuper},~\ref{var3}	  & 01 Jun. 11, 04 Jul. 02 	\\
HD 154090  & 154090 & 17 04 49.3 & --34 07 22.5       & B0.7Ia	& Cro06   &~\ref{obsuper}		  & 01 Jun. 09, 04 Jul. 02 	\\
HD 169454  & 169454 & 18 25 15.1 & --13 58 41.5      & B1Ia+    & Riv97    &\ref{obsuper},~\ref{var3}	  & 01 Jun. 09, 04 Jul. 02 	\\
\hline
&&&&&\\
\end{tabular}
\begin{tabular}{c c c c c c c c}
\multicolumn{8}{c}{Wolf-Rayet stars} \\
\hline\hline
Name & HD & RA (J2000) & Dec (J2000) & Spectral type & Reference & Fig. & Obs. Date(s)  \\
\hline
WR 6       & 50896  & 06 54 13.0 & --23 55 42.0  & WN4	   & vdH01    & ~\ref{wn}	 & 02 Nov. 04   \\
WR 11      & 68273  & 08 09 32.0 & --47 20 11.7  & WC8+O7.5III-V & vdH01    & ~\ref{wc}   & 01 Jun. 11   \\
WR 22      & 92740  & 10 41 17.5 & --59 40 36.9  & WN7h+O9III-V  & vdH01    & ~\ref{wn}   & 01 Jun. 09   \\
WR 78      & 151932 & 16 52 19.2 & --41 51 16.2  & WN7h    & vdH01    & ~\ref{wn}	 & 01 Jun. 09   \\
WR 90      & 156385 & 17 19 29.9 & --45 38 23.9  & WC7	   & vdH01    & ~\ref{wc}	 & 01 Jun. 11   \\
WR 103     & 164270 & 18 01 43.1 & --32 42 55.2  & WC9d     & vdH01    & ~\ref{wc}	 & 01 Jun. 11   \\
WR 136     & 192163 & 20 12 06.5 & +38 21 17.8  & WN6(h)     & vdH01	& ~\ref{wn}	 & 01 Jun. 09   \\
\hline
\end{tabular}
\end{table}